Author's response to reviews

Title: The difficult diagnosis of glioblastome multiforme in adults: case report and review of the literature

Authors:

Shaheen E Lakhan (slakhan@gnif.org)
Lindsey Harle (lharle@gnif.org)

Version: 4 Date: 24 December 2008

Dear Editor,

Thank you for conducting the peer-review. The only commentary for revision that we read pertained to the quality of the pathology images. However, the two images we uploaded adequately demonstrate the pons lesion with pleomorphic, poorly differentiated glial cells (especially on Figure 2, high power magnification). Moreover, these two are the only images that we can provide for publication. Therefore, the manuscript and images remain unchanged in this revision submission.

We look forward to seeing this case published in a timely manner. Thank you.

Sincerely,

Shaheen E Lakhan & Lindsey Harle